Author's response to reviews

Title: Prevalence of Prehypertension and its Relationship to Risk Factors for Cardiovascular Disease in Jamaica: Analysis from a Cross-Sectional Survey

Authors:

Trevor S Ferguson (trevor.ferguson02@uwimona.edu.jm)
Novie O Younger (novie.younger@uwimona.edu.jm)
Marshall K Tulloch-Reid (marshall.tullochreid@uwimona.edu.jm)
Marilyn B Lawrence Wright (marilyn.lawrencewright@uwimona.edu.jm)
Elizabeth M Ward (warde@moh.gov.jm)
Deanna E Ashley (deca@cwjamaica.com)
Rainford J Wilks (rainfordw@cwjamaica.com)

Version: 2 Date: 17 January 2008

Author's response to reviews:

References were duplicated in error. This has been corrected.